Author's response to reviews

Title: Duodenal Somatostatinoma: A Case Report and Review of Literature

Authors:

oguz koc (oguzcoach@yahoo.com)
yigit duzkoylu (dryigit@gmail.com)
yavuz s sari (yssari2003@yahoo.com)
hasan bektas (hasanbektas3417@hotmail.com)
gungor uzum (guzum@yahoo.com)
vahit tunali (vtunali@hotmail.com)
esra pasaoglu (epasa@yahoo.com)

Version: 4 Date: 27 October 2012

Author's response to reviews: see over
To Editorial Board:

Here, in the revised form of our report,

We describe our patient (Turkish white male) more specifically. Histopathological findings, lymph node status, the localization of the mass and our surgical procedure are explained in a more detailed form. We also cited the malignancy and the proliferation index of the tumor.

Yours sincerely,
Oguz Koc, M.D.
Corresponding Author